Author's response to reviews

Title: A repeated measurement study investigating the impact of school outdoor environment upon physical activity across ages and seasons in Swedish second, fifth and eighth graders

Authors:

Peter Pagels (peter.pagels@lnu.se)
Anders Raustorp (anders.raustorp@ped.gu.se)
Antonio Ponce De Leon (Antonio.Ponce.De.Leon@ki.se)
Fredrika Mårtensson (fredrika.martensson@slu.se)
Maria Kylin (maria.kylin@slu.se)
Cecilia Boldemann (cecilia.boldemann@ki.se)

Version: 3
Date: 17 July 2014

Author's response to reviews:

Dear Editor,

We have now asked a native English speaking colleague to edit the article “A repeated measurement study investigating the impact of school outdoor environment upon physical activity across ages and seasons in Swedish second, fifth and eighth graders”. A slight change has been made in the heading: the very last word “graders” replaces “grade pupils”, as suggested by the colleague. All corrections have been entered by her, and we hereby resubmit the article. We dispensed with a point-by-point response as the article has undergone a complete linguistic makeover, with a lot of minor and some major corrections (phraseology, grammar etc.). A point-by-point response would thus look too chaotic. We hope that this is ok.

Sincerely,

Peter Pagels